Author's response to reviews

Title: Pattern of Respiratory Diseases in Children Presenting to the Pediatric Emergency Unit of the University of Nigeria Teaching Hospital, Enugu: A Case Series Report

Authors:

Tagbo Oguonu Dr. (tagbo.oguonu@unn.edu.ng)
Adaeze C Ayuk Dr. (adaraymond@yahoo.com)
Benedict O Edelu Dr. (onedelu@yahoo.com)
Ikenna K Ndu Dr. (ikennandu@gmail.com)

Version: 2
Date: 2 January 2014

Author's response to reviews:

Dear Sir,

In response to the request that the Abstract should be formatted in line with the instructions to the authors. We have attached a revised copy of the document for your review. It is our hope that this will meet the requirement.

Best regards

Tagbo Oguonu

Corresponding author